Author's response to reviews

Title: Reporting of setting and health worker cadre among trials on antenatal iron and folic acid supplementation in pregnancy: a quality assessment based on two Cochrane reviews

Authors:

Rachel A Harper (rachel_harper@med.unc.edu)
Simon Lewin (simon.lewin@kunnskapssenteret.no)
Claire Glenton (claire.glenton@kunnskapssenteret.no)
Juan P Pena-Rosas (penarosasj@who.int)

Version: 2 Date: 12 March 2013

Author's response to reviews:

March 12, 2013

Dear Editor-in-chief:

Included below are a list of the revisions made to our manuscript “Reporting of setting and health worker cadre among trials on antenatal iron and folic acid supplementation in pregnancy: a quality assessment based on two Cochrane reviews” by Harper R, et al.

1) Appropriate title page included with title, author names, author affiliations, author addresses and email addresses
2) The abstract has been included in the main manuscript on page 2
3) Table 1 has been included on page 7 of the primary manuscript
4) The completed PRISMA checklist and flow diagram are attached as separate documents

We appreciate your initial review and suggestions.

Thank you,

Rachel Harper
Associate Researcher
Norwegian Knowledge Centre for Health Services